Figure 3.18. OTHER GRAPHICAL METHODS: Napoleon’s Russian Campaign [A diagram adapted and redrawn from a reproduction of the French original (1861)]

Ne m a n R.

ILLUSTRATIVE MAP of

6,000 22,000

Kaunas Vilnius

Smorgon’

−38o 6 Dec.

−32o 7 Dec.

successive losses of men from the French Army in the Russian Campaign, 1812-1813

Prepared by M. Minard, retired Inspector General of Bridges and Dykes •

Polotsk

Glubokoye

Molodechno

Minsk

Studienka

Borisov

Vitebsk

Orsha

Mogilev

Smolensk

Dorogobuzh

Gzhatsk

Vyaz’ma

Mozhaysk

MOSCOW

Tarutino

Maloyaroslavets

Be rez ina R. Dn e p r R.

Mo sk va R.

Kilometres (approximate)

French common leagues

0 57 114 171

0 25 50 75

GRAPH of the temperature in^oC below zero during the retreat from Moscow

0o 18 Oct. 10 20 30 40

Rain 24 Oct.

−11o 9 Nov.

−26o 14 Nov.

−14o

−25o 28 Nov. −30o 1 Dec.

NOTES: 1. The English spelling of some names on the map varies in different sources; the spellings used above are those in Plates 45 and 46 of The Times Atlas of the World, Comprehensive Edition, Times Books, New York, 1980, ISBN 0-8129-0906-2. [DC Library call number G1021.T551980]

2. Although the distance of one league has differed both with country and with time, a metric equivalent of about 4 km would be expected in this map; thus, either the scale 0, 25, 50, 75 is wrong (0, 15, 30, 45 would be closer to correct) or a different distance for the French common league is involved.

University of Waterloo STAT 220 – W. H. Cherry

#3.37

100,000

100,000

1995-04-20 (continued overleaf )

10,000 4,000

8,000

14,000

12,000

28,000

50,000

20,000

24,000

37,000

55,000

87,000

96,000

422,000

400,000

60,000

33,000

175,000

145,000

127,100

100,000